Author’s response to reviews

Title: Age at Menarche and Risk of Gestational Diabetes Mellitus: A Population-Based Study in Xiamen, China

Authors:

xuejun li (xmlxuejun@163.com)
Liyang Wang (wlywly1979@163.com)
bing Yan (yanbing.sy@163.com)
xiulin Shi (shixiulin2002@163.com)
haiqu Song (haiqs@163.com)
weijuan su (swj851016@126.com)
bingkun Huang (bk.huang@outlook.com)
yuxian Zhang (304682726@qq.com)
shunhua Wang (850387761@qq.com)
fuping lv (lvfp1989@126.com)
mingzhu Lin (linmz65@tom.com)

Version: 4 Date: 13 Mar 2019

Author’s response to reviews:

1. We note that you have selected "Database" as your article type. However, based on our submissions guidelines, we would like to request that you change your article type to "Research Article."

Thank you very much for your expert comments. We had revised our paper following your suggestion.
2. Please ensure that the authors list is the same in both the submissions system and on the title page, as we note that the corresponding author is included as the first author instead of the last author in the submission system.

Thank you very much for your expert comments. We had revised our paper following your suggestion.

3. We note that WS, BH, YZ, SW, and FL are missing from the Authors’ Contributions section, but are included on the title page. We also note that YY, ZL, PH, and SY are included in the Authors’ Contributions section, but it is unclear whose initials they correspond with on the title page. Therefore please ensure that all authors included in the authors list are also included in this section. Please also include the authors initials in the order of the authors given name and then the authors family name ( ex. Xuejun Li, would be XL).

Thank you very much for your expert comments. We had revised our paper following your suggestion.

4. If the need for informed consent has been waived by an IRB or is deemed unnecessary according to national regulations, please clearly state this with details, including the name of the Board or a reference to the relevant legislation in the ‘Ethics approval and consent to participate’ section of the Declarations.

Thank you very much for your expert comments. The informed consent did not be revised.

5. Please remove the publisher's note from the Declarations section.

Thank you very much for your expert comments. We had revised our paper following your suggestion.
6. Please proofread ensure that when you upload your revised submission that it is in the final form for publication. Please remove any tracked changes, colored text, or highlighting and include only a single clean copy of the manuscript. Should you wish to respond to these revision requests, please include the information in the designated input box only.

Thank you very much for your expert comments. We had revised our paper following your suggestion.